\begin{table} %t1
\caption{\label{pulsartable1}Light curve classification of 60 X-ray pulsars observed at\protect energies above 10~keV.}
%\centering
\par
\begin{tabular}{lcccc}
\hline\hline\noalign{\smallskip}
Name & $P_{\rm spin}$\tablefootmark{a}  	&  Pulses\tablefootmark{b}  &	B03\tablefootmark{c} & Ref. \\ % 
\hline\noalign{\smallskip}
 RX J0051.8--7310 		& 16.6&1 && 1 \\
RX J0052.1--7319 	 	& 15.4 & 2 & 2 & 2\\
XTE-SMC95 			& 95	&1 & & 3 \\
SMC X-2 	 		& 2.37 & 1 & 1 & 4 \\
XTE J0055--724  		& 59	& 1 & 1 & 5 \\
SMC X-1 	 		& 0.71 &2&2& 6 \\
RX J0117.6--7330 		& 22	& 2 & 2 & 7\\
2S 0114+650   		& 9828 &1& 1 & 8\\
4U 0115+63  		& 3.6  &1 & 1 & 9, 16  \\
4U 0142+614\tablefootmark{d} & 8.7 & 2 & 2 & 10 \\
RX J0146.9+6121    	& 1408 & 1 & 1 & 11 \\
V 0332+53 	 		& 4.37  & 2 & 1 & 12 \\
4U 0352+309 = X Per   	& 835   & 1 & 1 & 13 \\
EXO 053109--6609.2   	& 13.7  & 2 &    & 14 \\
LMC X-4            		& 13.5  & 1 & 1 & 15 \\
1A 0535+26   		& 105   & 2 & 2 & 16 \\
MXB 0656--072 		& 160   & 1 &    & 17 \\
4U 0728--25  		& 103   & 2 & 2 & 18 \\ % X0726--260=
RX J0812.4--3114   	& 31.9  & 2 & 2 & 19 \\ % LS 992=
GS 0834--430       		& 12.3  & 2 & 2 & 16, 20 \\
Vela X-1 		  	& 283 & 2& 2 & 16, 21 \\
GRO J1008--57  	 	& 93.5 & 1 & 1 & 16 \\
1A 1118--616   		& 405& 1 & 1&16, 22 \\
Cen X-3 	 		& 4.82 & 2 &	2 &   16, 23 \\
1E 1145.1--6141    	& 297 & 2 & 1 &  16, 21 \\
4U 1145--619 		& 292 & 1 & 1 &         16 \\
GX 301--2   			& 681 & 2 & 2 & 16, 24 \\
GX 304--1   			& 272 & 1(flat) & 1 & 25 \\
2S 1417--624 	 	& 17.6 & 2 & 2 & 16  \\
4U 1538--52 	 	& 530 & 2 & 2 & 16, 21 \\
XTE J1543--568 		& 27.1 & 2 &    & 26  \\
SWIFT J1626.6--5156 	& 15.4 & 1 & & 27 \\
IGR J16358--4726 		& 228 &1 & & 28 \\
IGR J16393--4643 		& 912 &2 & & 29 \\
OAO 1657--415 	  	& 37.7 & 1 & 1 & 16 \\ 
1RXS J170849.0--400910\tablefootmark{d}  & 11.0  &1 &1 & 10  \\
GPS 1722--363  		& 414 & 1 & 1 & 30  \\  
AX J1749.1--2733 		& 132 &2 & & 31  \\
GRO J1750--27 	 	& 4.45 & 1 &	1 &   16,  32 \\
SAX J1802.7--2017 	& 140 &2 & & 33 \\
SGR 1806--20\tablefootmark{d}   & 7.47 &2 & & 34  \\
Sct X-1 		 	& 111 & 2 &	2 & 35 \\  
GS 1843+00 	 	& 29.5 & 2 & 2 &         36 \\
GS 1843--024  	 	& 94.3 & 1  & 1 &     37 \\
IGR J18483--0311 		& 21.1 & 2 & & 38  \\
XTE J1855--026  	 	& 361 & 1	& 1 & 39 \\
XTE J1858+034 	 	& 221 & 1 & 1 &    40  \\
SGR 1900+14\tablefootmark{d}  & 5.16 &2 & & 34 \\
4U 1901+03 		& 2.76 &2 & &41 \\
XTE J1906+09  	 	& 89 & 2 & 1 &      42 \\
4U 1907+09   		& 440 & 2 & 2 & 43 \\
4U 1908+075 		& 605 &2 & & 44 \\
XTE J1946+274     		& 15.8 & 2 & 2 &      45 \\
KS 1947+300 		& 18.7 & 1 &  1 & 46, 47 \\ 
\hline
SW J2000.6+3210 		& 1056 & 1 & & 48  \\
EXO 2030+375 	 	& 42 & 2 & 2 &       16,  49 \\
GRO J2058+42 	 	& 198 & 1 &	1 &    16  \\
SAX J2103.5+4545		& 359 & 1 & 1 & 50 \\
Cep X-4  		 	& 66.2 & 2 & 2 & 51  \\
1E 2259+586\tablefootmark{d}  & 6.98 & 2 & 2 &  52 \\
\hline
\end{tabular}
\tablefoot{\tablefoottext{a}{Pulsar spin period (s).}
\tablefoottext{b}{Number of pulses in the profile.}
\tablefoottext{c}{Classification by B03.}
\tablefoottext{d}{Magnetars.}}
\tablebib{(1)~\citet{2002ApJ...567L.129L}; (2)~\citet{2001ApJ...560..378F};
(3)~\citet{2002AA...385..464L}; (4)~\citet{2001ApJ...548L..41C}; 
(5)~\citet{1998AAA...338L..59S}; (6)~\citet{1993ApJ...410..328L}; 
(7)~\citet{1999ApJ...518L..99M}; (8)~\citet{2000ApJ...536..450H};
(9)~\citet{1999ApJ...523L..85S}; (10)~\citet{2006ApJ...645..556K};
(11)~\citet{2000AAA...354..567M}; (12)~\citet{2006MNRAS.371...19T};
(13)~\citet{1989ApJ...346..469R}; (14)~\citet{1998ApJ...498..831B}; 
(15)~\citet{1996ApJ...467..811W}; (16)~\citet{1997ApJS..113..367B};
(17)~\citet{2006AA...451..267M}; (18)~\citet{1997ApJ...489L..83C}; 
(19)~\citet{1999MNRAS.306...95R}; (20)~\citet{1992PASJ...44..641A};   
(21)~\citet{1995PhDT.......215M}; (22)~\citet{1994AAA...289..784C};
(23)~\citet{2000ApJ...530..429B}; (24)~\citet{1997ApJ...479..933K};
(25)~\citet{1977ApJ...216L..15M}; 
(26)~\citet{2001ApJ...553L.165I}; (27)~\citet{2008AA...485..797R};
(28)~\citet{2005AA...444..821L}; 
(29)~\citet{2006AA...447.1027B}; (30)~\citet{1989PASJ...41..473T}; 
(31)~\citet{2008MNRAS.386L..10K}; (32)~\citet{1997ApJ...488..831S};
(33)~\citet{2003ApJ...596L..63A}; (34)~\citet{2002ApJ...577..929G};
(35)~\citet{1991ApJ...370L..77K};
(36)~\citet{1990ApJ...356L..47K}; (37)~\citet{1999ApJ...517..449F};
(38)~\citet{2007AA...467..249S}; (39)~\citet{1999ApJ...517..956C};
(40)~\citet{1998AetA...337..815P}; (41)~\citet{2005ApJ...635.1217G};
(42)~\citet{2002ApJ...565.1150W}; (43)~\citet{1998ApJ...496..386I};
(44)~\citet{2004ApJ...617.1284L}; (45)~\citet{2003ApJ...584..996W};
(46)~\citet{2005AstL...31...88T}; (47)~\citet{1995ApJ...446..826C}; 
(48)~\citet{2009ApJ...699..892M};  (49)~\citet{1999ApJ...512..313S};
(50)~\citet{1998AAA...337L..25H}; (51)~\citet{1991ApJ...366L..19K}; 
(52)~\citet{1992PASJ...44....9I}.
}
\end{table}